Author’s response to reviews

Title: Cost And Effectiveness Of Prescribing Emollient Therapy For Atopic Eczema In UK Primary Care in Children and Adults: A Large Retrospective Analysis Of The Clinical Practice Research Datalink

Authors:

George Moncrieff (melissa.mandell@envisionpharmagroup.com; georgemoncrieff@hotmail.com)

Annie Lied-Lied (ALIEDLIE@ITS.JNJ.com)

Gill Nelson (gnelson1@ITS.JNJ.com)

Chantal Holy (CHoly1@its.jnj.com)

Rachel Weinstein (RWeinst1@its.jnj.com)

David Wei (dwei8@ITS.JNJ.com)

Simon Rowe (simon.m.rowe@gmail.com)

Version: 1 Date: 19 Jun 2018

Author’s response to reviews:

Comment #1 has been addressed in the manuscript and the first paragraph of the Methods section was slightly amended.

Comment #2 was also addressed and the revised paper does not have any track changes.